
NSF CC-NIE Working Group Meetng 

March 24, 2014 

Agenda 

 

Time: 3:30-4:30pm, March 24, Monday 

Locaton: Physics building, Conference room 403B  

Agenda: 

1. Status of the monitoring infrastructure 

2. Update from CEN  

3. Status of Science-DMZ equipment at Storrs 

4. Update from UCHC  

5. Next step 

6. Open discussion on other issues 

Minutes  
Present: 

UConn: Bob, Bing, Jef F., Richard (Jones)  

UCHC: Jef Duton, Sophan, Leo 

1. Status of the monitoring infrastructure  

Bing went over the throughput analysis results with Levon. Levon is in the process of incorporatng the 
comments.    

2. Update from CEN 

CEN is stll looking at an April 14th delivery for the DWDM equipment.  The vendor is engaged and 
wishes to assist us with the turn-up of the equipment mid-April.  

3. Status of Science-DMZ equipment at Storrs 

Jef, Bob, Bing and Richard met last week to discuss how to connect Richard’s lab to the Science DMZ. 
Bob will look into several candidate devices. 

   



4. Update from UCHC 

Jef D. has been working with Richard and made good progress on the OSG site. UCHC has all gear on 
hand for the 100/40GbE backend except some minor cables.  

5. Next step 

Richard mentons that his team has started the data challenge. He has not rolled out the experiments 
because of power outage in Physics building, but plan to start by Wednesday. Jef will monitor the 
network trafc correspondingly.   
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